Author's response to reviews

Title: Outcomes of Total Hip Arthroplasty in Patients with Primary Immune Thrombocytopenia

Authors:

Seung-Jae Lim (limsj70@gmail.com)
Ingwon Yeo (igyeo@samsung.com)
Chan-Woo Park (cwpark@samsung.com)
Young-Wan Moon (ywmoon@skku.edu)
Youn-Soo Park (ysp3504@skku.edu)

Version: 8  Date: 16 August 2015

Author's response to reviews:

Date: August 16, 2015

Responses to Editor’s Comment

We would like to express our sincere thanks to the editor for providing us a valuable comment.

Response to editor’s comment:

According to our editorial policies, if 3 or more indirect identifiers are provided for any participant, consent to publish this information is required. Table 2 and the legend for figure 1 contain indirect identifiers, making consent to publish necessary. Please confirm whether consent to publish was obtained, clearly stating this in the manuscript. If not, please remove the identifiers from the figure legend and collate the information in table 2 so that identifiers for individual participants are not provided, or remove both the table and figure.

Response: Thank you for your comment. We have clearly stated this information in the manuscript as shown below;

Page 6, lines 74-76

“This study protocol was approved by the Review Board of Samsung Medical Center. Written informed consent for participation in the study was obtained from all patients.”